Author's response to reviews

Title: Traumatic bone cyst of the mandible of possible iatrogenic origin. A case report and brief review of the literature.

Authors:

Arsinoi A Xanthinaki (xanthinaki@yahoo.gr)
Konstantinos I Choupis (choupis@hol.gr)
Konstantinos Tosios (ktosios@dent.uoa.gr)
Vasilios A Pagkalos (bapale@med.uoc.gr)
Stavros I Papanikolaou (iritzou@dent.uoa.gr)

Version: 2 Date: 7 November 2006

Author's response to reviews:

Dear Madam or Sir:

Subject: Submission of revised manuscript

Thank you for accepting our manuscript for publication. In reply to your email we are sending you a revised version of the manuscript. The changes that have been made are:

1. All unnecessary capitals have been removed from the title.
2. The text: "title" and "authors" have been removed from the title page.
3. The authors’ qualifications have been removed.
4. The affiliations have been written in the requested format.
5. The corresponding author and her email address have been reported on the title page.
6. Keywords have been removed from the manuscript.
7. The "Main text" has been renamed to "Background".
8. References have been placed before the full stop.
9. 3 commas have been added to the text (pages 2 and 4).
10. The text "have rarely been reported" (page 5) has been replaced by "are rather uncommon in the literature".

Yours,

Arsinoi A Xanthinaki DDS, MD

Correspondence:
email: xanthinaki@yahoo.gr
Arsinoi A Xanthinaki DDS, MD
Tetrapoleos 14,
11527, Athens, Greece
tel +30-2107784492
tax +30-2810234756